Author's response to reviews

Title: Use of the GenoType(R) MTBDRplus assay to assess anti-TB drug resistance levels and patterns among Mycobacterium tuberculosis isolates from TB patients in rural Uganda

Authors:

Joel Bazira (jap2k2001@yahoo.co.uk)
Benon B Asiimwe (basiimwe@chs.mak.ac.ug)
Moses Joloba (moses.joloba@case.edu)
Fred Bwanga (fbwanga@chs.mak.ac.ug)
Mecky I Matee (mmateemi@yahoo.com)

Version: 2 Date: 2 February 2010

Author's response to reviews:

Line 207 has changed from "However there is evidence of an increase in the" to However there is evidence of an increase in the MDR rate in Uganda in the last two studies compared to the first National survey.